Author's response to reviews

Title: Postnatal depressive symptoms amongst women in Vietnam: A cross-sectional study investigating prevalence and associations with social, cultural and infant factors

Authors:

Linda J Murray (linda.murray@utas.edu.au)
Michael P Dunne (m.dunne@utas.edu.au)
Thang Vo Van (vovanthang147@gmail.com)
Nguyen Thi An Phuong (thuynguyen201@gmail.com)
Nigar Khawaja (n.khawaja@qut.edu.au)
Thanh Cao Ngoc (thuynguyen201@gmail.com)

Version: 6 Date: 27 August 2015

Author's response to reviews:

Dear Sir/Madam,

Thank you for your favourable review of our manuscript. As per your request we have made the following final edits:

- A statement about informed consent has been added to the Ethics section
- The email address for all authors have been added, and the contact details and names for all authors have been updated
- Funding is now included in the Acknowledgements section, and the separate funding section has been deleted.
- The section on competing interests has been re-named "Competing interests"
- The section on Author’s contributions has been re-named "Author’s contributions"

Please don’t hesitate to contact us if you have any further questions about this manuscript.

Regards

Dr Linda Murray